Author's response to reviews

Title: A Neonatal Presentation Of Factor V Deficiency: A Case Report

Authors:

  Amol Chingale (dramolchingale@hotmail.com)
  Michael Eisenhut (michael_eisenhut@yahoo.com)
  Anjali Gadiraju (rajuanju1@yahoo.co.in)
  Ri Liesner (liesnr@gosh.nhs.uk)

Version: 4 Date: 22 January 2007

Author's response to reviews:

Reply to the reviewers report.

We have shortened the manuscript and corrected the typographical errors. A comprehensive listing of details of all cases in journals of all languages would be important for a systematic review but is not essential for a simple case report. Other reports of patients with factor V deficiency appeared in journals in Hungarian, Dutch, Swedish, Spanish and Italian language. Due to lack of sufficient knowledge of these languages and funding for a professional translator as well as retrieval of the papers we could not include these cases. We have included the persons name heading the laboratory were the genetic testing was done in the acknowledgements. We have omitted the picture of the nipple.
Thank you for your thorough review and helpful suggestions.

Yours sincerely

On behalf of all authors

Michael Eisenhut